Author's response to reviews

Title: Modification of additive effect between vitamin and ETS on childhood asthma risk by GSTP1 polymorphism: a cross sectional study

Authors:

So-Yeon Lee (imipenem@hanmail.net)
Bong-Seong Kim (bskped@hanmail.net)
Sung-Ok Kwon (jaia0609@khidi.or.kr)
Se-Young Oh (seyoung@khu.ac.kr)
Hye Lim Shin (ysb1126@hanmail.net)
Young Ho Jung (youngdr1@naver.com)
Eun Lee (unelee@hanmail.net)
Song-I Yang (qjit222@naver.com)
Hyung Young Kim (hawmok77@hanmail.net)
Ju-Hee Seo (juheesea@hanmail.net)
Byoung-Ju Kim (drkbj@hanmail.net)
Hyo-Bin Kim (hbkim1126@naver.com)
Ji-Won Kwon (pedas@naver.com)
Hae-Ran Lee (drran@hallym.or.kr)
Soo-Jong Hong (sjhong@amc.seoul.kr)

Version: 2
Date: 13 August 2014

Author's response to reviews:

Thank you for your email. As you comment, I corrected our manuscript. Thank you.